Author’s response to reviews

Title: Pharmacological potential of Bidens pilosa L. and determination of bioactive compounds using UHPLC-QqQ-LIT-MS/MS and GC/MS

Authors:
Garima Singh (garima.singh106@gmail.com)
Ajit Passari (ajit.passari22@gmail.com)
Vincent Leo (vincentvineethleo@gmail.com)
Sarathbabu Subbarayan (sarah07bio@gmail.com)
Pratibha Singh (garimasingh@mzuonline.in)
Brijesh Kumar (gbrikum@yahoo.com)
Bhim Singh (bhimpratap@gmail.com)
Hauzel lalhlenmawia (hlenmawia@gmail.com)
Nachimuthu Senthil kumar (nskmzu@gmail.com)

Version: 1 Date: 28 Oct 2017

Author’s response to reviews:

Authors responses to reviewer(s) comments

Authors are very much thankful to all the reviewers and editor for their valuable comments. We have revised manuscript as per your suggestions and incorporated the correction in the revised manuscript. Please find below the answers and rebuttals of the queries raised by editor/reviewers.

Reviewer #1

Reviewer 1 has not asked for any changes in the manuscript.

Reviewer #2

Reviewer’s comment: 1.
Page 9; Line 202-203:- Authors can explain in brief the method obtained.

Author’s response: 1.

Lines 202-203 has been modified as per suggestion of reviewer as “A mixed standard stock solution (1 mg/mL) of five reference compounds was prepared in methanol. The working standard solutions were prepared by appropriate dilution of the mixed standard solution with acetonitrile to a series of concentration ranges from 0.1-1000 ng/mL. The standard stock and working solutions were stored at -20°C until use and vortexed for 30 s prior to injection.”

Reviewer’s comment: 2.

Did authors obtained any derivatization steps during pre-treatment of samples before its analysis.

Author’s response: 2.

No derivatization obtained

Reviewer’s comment: 3.

Authors did not marked any significant results (*) in the tables and figures. If possible please include.

Author’s response: 3.

Table 2 and Table 3 significant values has been marked as suggested by the reviewer.